Author's response to reviews

Title: Audit of therapeutic interventions in inpatient children using two scores: are they evidence-based in developing countries?

Authors:

Nilton Y Carreazo (yhuro@hotmail.com)
Carlos A Bada (ifocar@yahoo.com)
Juan P Chalco (jpcho33@yahoo.com)
Luis Huicho (lhuicho@viabcp.com)

Version: 8 Date: 26 December 2004

Author’s response to reviews:

Lima, December 26, 2004

Dr. Alison Z Shaw, Ph.D.
Assistant Editor
editorial@biomedcentral.com

Dear Dr. Shaw,

Many thanks for your most recent correspondence dated December 21, 2004.

As you suggested, we constructed a new table and renumbered it and the subsequent ones accordingly. Thus, Table 4 (continued) from the former version is Table 5 now, Table 5 is Table 6 and Table 6 is Table 7. These changes are also referred to in the text.

We are re-submitting a new version of the manuscript now.

Sincerely,

Luis Huicho